Author's response to reviews

Title: Factors influencing the implementation of a lifestyle counseling program in patients with venous leg ulcers: A multiple case study

Authors:

Irene M van de Glind (i.vandeglind@iq.umcn.nl)
Maud M Heinen (m.heinen@iq.umcn.nl)
Andrea W Evers (a.evers@mps.umcn.nl)
Michel Wensing (m.wensing@iq.umcn.nl)
Theo van Achterberg (t.vanachterberg@iq.umcn.nl)

Version: 5 Date: 16 October 2012

Author's response to reviews: see over
Dear editor

Please find attached the clean version of the manuscript. I have accepted all proposed changes.

Please could you check the font of the Manuscript Title, Reference list and Tables/Figures? I think it still is in Arial while the rest of the manuscript is in Times New Roman.

There were some questions from the copy-editor in the manuscript. I will answer these below.

If there are any questions, please let me know.

With kind regards,

Irene van de Glind

Questions from the copy-editor (4 questions):

1) Authors first names are:
   Irene M. van de Glind§, Maud M. Heinen¹, Andrea W. Evers², Michel Wensing¹, Theo van Achterberg¹

2) I have added in this reference to Figure 1, as per style guidelines, articles are not to have suggested placements for figures. Please confirm this mention is OK.
   This is ok.

3) Tables should be referred to in order. Table 2 is currently referred to before Table 1. Either add a Table 1 reference before this mention or renumber the tables.
   I've added a reference to Table 1 on page 7.

4) Per IS style guidelines, all figures should be accompanied by a figure legend that begins with a short title (max 15 words). Please provide such a legend.
   On page 31 we added the following legend:

   A framework to evaluate the implementation of a lifestyle program. This framework, adapted from Hasson 2010, describes different areas that could influence the outcome of a lifestyle program: Influencing factors (e.g. Implementation strategies and delivery, Patient recruitment, Participant responsiveness, Organizational preconditions, Knowledge/support of colleagues, Context), Program adherence and Implementation success.